Towards identifiability of micro total effects in summary causal graphs with latent confounding: extension of the front-door criterion

charles.assaad@inserm.fr

Charles K. Assaad Sorbonne Université, INSERM Institut Pierre Louis d'Epidémiologie et de Santé Publique F75012, Paris, France

Reviewed on OpenReview: https://openreview.net/forum?id=5f7YlSKG11

Abstract

Conducting experiments to estimate total effects can be challenging due to cost, ethical concerns, or practical limitations. As an alternative, researchers often rely on causal graphs to determine whether these effects can be identified from observational data. Identifying total effects in fully specified causal graphs has received considerable attention, with Pearl's frontdoor criterion enabling the identification of total effects in the presence of latent confounding even when no variable set is sufficient for adjustment. However, specifying a complete causal graph is challenging in many domains. Extending these identifiability results to partially specified graphs is crucial, particularly in dynamic systems where causal relationships evolve over time. This paper addresses the challenge of identifying total effects using a specific and well-known partially specified graph in dynamic systems called a summary causal graph, which does not specify the temporal lag between causal relations and can contain cycles. In particular, this paper presents sufficient graphical conditions for identifying total effects from observational data, even in the presence of cycles and latent confounding, and when no variable set is sufficient for adjustment.

1 Introduction

Causal questions arise when we seek to understand the effects of interventions, such as asking, "If we administer today a hypertension treatment to a patient with kidney insufficiency, will the kidney function (represented by the creatinine level) improve tomorrow?". These questions, often referred to as total causal effects or simply total effects (Pearl et al., 2000), are denoted¹ as $Pr(Creatinine_{tomorrow} = c \mid do(Hypertension_{today} = h))$ where the do() operator denotes an intervention. They differ from associational relationships, $Pr(Creatinine_{tomorrow} = c \mid Hypertension_{today} = h)$, as they isolate the effect of an intervention, disregarding other influencing factors such as sodium intake or protein intake or stress level. Experimentation is known as the traditional approach across various fields to estimate the total effect of interventions free from confounding bias (Neyman et al., 1990). However, conducting experiments is not always feasible due to cost, ethical considerations, or practical limitations. Consequently, scientists often resort to estimating effects of interventions from observational data. This process relies on specific assumptions and typically involves two sequential steps: identifiability and estimation (Pearl, 2019). The identifiability step refers to the question of whether the total effect of interest can be uniquely determined from the available data and the assumptions made about the causal model. This step usually also include finding a way to express the total effect in terms of the observed data distribution, i.e., using a do-free expression. On the

¹In a nonparametric setting, the total effects is a specific functional of $Pr(Creatinine_{tomorrow} = c \mid do(Hypertension_{today} = h))$ for different values of h. However, for simplicity, the total effect is often reffered to as $Pr(Creatinine_{tomorrow} = c \mid do(Hypertension_{today} = h))$.

other hand, the estimation step refers to the process of calculating the value of a total effect, once it has been identified, from finite observational data using statistical methods. This paper focuses on the first step.

Graphical models, provide a framework for identifying total effects from causal graphs which encode variables as vertices and causal relationships as arrows, allowing researchers to visualize and analyze complex causal structures (Pearl et al., 2000). Identifying total effects in fully specified non-temporal causal graphs has been a subject of considerable attention (Pearl, 1993b; 1995; Spirtes et al., 2000; Pearl et al., 2000; Shpitser & Pearl, 2008; Shpitser et al., 2010). Adjusting for covariates is one method among several that allow us to identify causal effects and the back-door criterion (Pearl, 1993b) is one of the most known methods that allow us to find covariates using causal graphs. Pearl (1993a; 1995) has provided examples where no set of variables is sufficient for adjustment, yet the causal effect can still be consistently estimated through multi-stage adjustments. Pearl's front-door criterion offers a graphical method for identifying total effects despite the absence of an adjustment set due to latent confounding, assuming the causal graph is a directed acyclic graph (DAG) or an acyclic directed mixed graph (ADMG). Initially criticized for its limited practical application, this criterion has recently gained recognition and is now employed in epidemiology (Inoue et al., 2022; Piccininni et al., 2023).

The above identifiability methods are directely applicable to fully specified temporal graphs (Blondel et al., 2016) which represent causal relations in dynamic systems where causal relationships evolve over time. However, constructing a fully scpecified temporal graph requires knowledge of all causal relationships among observed variables, which is often unavailable, especially in many real-world applications. However, experts may know that one variable causes another without knowing the exact temporal lag. For example, understanding the transmission of SARS-CoV-2 from younger to older individuals, and vice versa, can help define interventions most likely to reduce the number of deaths. Indeed, it has been shown that younger adults tended to be highly infected during the first wave of the pandemic, while older individuals faced a higher risk of death if infected (Carrat et al., 2021; Lapidus et al., 2021; Glemain et al., 2024). Considering sufficiently large time intervals (several weeks) as in repeated serosurveys, like in Wiegand et al. (2023), it is not clear if the number of new infections in one age group during a time interval (incidence) may be influenced by incidence in the other age group during the same interval. Incidence in an age group can also be influenced by incidence during the previous time interval in any age group. Therefore, constructing a fully specified causal graph is difficult. In such cases, partially specified causal graphs can be useful. A very well known and useful partially specified causal graph is the summary causal graph (SCG) which represents causal relations without including temporal information, i.e., each vertex represents a time series. Both medical and epidemiological examples given above can be represented by an SCG with a cyclic relationship representing the interplay between creatinine and hypertension in the first example and between the two age groups in the second example.

Recently, there has been new interest in extending identifiability results to partially specified graphs Eichler & Didelez (2007: 2010); Maathuis & Colombo (2013); Perkovic (2020); Wang et al. (2023); Anand et al. (2023); Ferreira & Assaad (2024); Assaad et al. (2024); Ferreira & Assaad (2025). Most of these partially specified graphs represent Markov equivalence classes, where the partial specifications differ conceptually from those in SCGs. For instance, in these graphs, partial specification typically manifests as undirected edges or edges with specific endpoints indicating uncertainty about the orientation. Additionally, each vertex in these graphs corresponds directly to an observed variable, maintaining a straightforward one-to-one relationship. As a result, the extensions of identifiability results for these graphs (Maathuis & Colombo, 2013; Perkovic, 2020; Wang et al., 2023) are not applicable to SCGs. Another important type of partially specified graph is the cluster graph, which represents causal relationships between clusters of variables rather than individual variables. This means that, unlike graphs representing Markov equivalence classes, the skeleton of a cluster graph does not correspond to the skeleton of true causal graph. SCGs are a specific type of cluster graphs, where each cluster represents a time series. Most works extending identifiability results to cluster graphs have focused on extreme multivariate cases, where the goal is to identify the total effect of one entire cluster on another entire cluster (Anand et al., 2023; Ferreira & Assaad, 2025). The few studies that consider the total effect of a single variable within a cluster on another single variable within a different cluster have been conducted on SCGs. For example, under the assumptions of no instantaneous relations and no latent confounding between two timepoints within the same time series, Eichler & Didelez (2007; 2010)

demonstrated that both the back-door and front-door criteria are applicable in SCGs. Assaad et al. (2023) further established that in the absence of latent confounding, the total effect remains identifiable through adjustment in acyclic SCGs, even when instantaneous relations are present. More recently, Assaad et al. (2024) provided graphical conditions for identifying total effects via adjustment in SCGs that incorporate both cycles and instantaneous relations, still under the assumption of no latent confounding. Finally, independently and concurrently of this work, Reiter et al. (2024) investigated the identification of total effects in the frequency domain using the SCG (referred to as a process graph in their work) under the assumption of linearity; however, unlike this and previous works (Eichler & Didelez, 2007; 2010; Assaad et al., 2023; 2024), they did not explore specific graphical conditions within the SCG for identification.

The previous paragraph underscores the novelty of this work: in the setting where a fully specified temporal causal graph is not available, this paper is the first to address the challenge of nonparametrically and graphically identifying total effects of *one variable within a cluster* on *another variable in a different cluster* when having access to an SCG while allowing *instantaneous relations*, *cycles*, and *latent confounding* (even between two timepoints within the same time series). It shows that the standard front-door criterion (Pearl, 1993a; 1995) and its previous extension to time-series (Eichler & Didelez, 2007; 2010) are not sound when applied to SCGs in the general setting considered in this paper. Nevertheless, by leveraging the front-door criterion, this work introduces sufficient conditions to identify total effects from observational data using SCGs in scenarios where latent confounding prevents identifiability through standard adjustment methods.

The remainder of the paper is organized as follows: Section 2 introduces necessary terminology and tools and it formalizes the problem. Section 3 demonstrates that the standard front-door criterion and its previous extention to time series are unsuitable when applied to SCGs. Section 4 presents the main result of this paper and Section 5 discusses several examples of non-identifiability. Finally, Section 6 concludes the paper.

2 Preliminaries and problem setup

This section, first introduces some terminology and tools which are standard for the major part and then, formalize the problem that this paper is going to solve. Let us start by defining the causal model that is considered.

Definition 1 (Discrete-time dynamic structural causal model (DTDSCM)). A discrete-time dynamic structural causal model is a tuple $\mathcal{M} = (\mathbb{L}, \mathbb{V}, \mathbb{F}, P(\mathbb{I}))$, where $\mathbb{L} = \bigcup \{\mathbb{L}^{v_t^i} \mid i \in [1, d], t \in [t_0, t_{max}]\}$ is a set of exogenous variables, which cannot be observed but affect the rest of the model. $\mathbb{V} = \bigcup \{\mathbb{V}^i \mid i \in [1, d]\}$ such that $\forall i \in [1, d], \mathbb{V}^i = \{V_t^i \mid t \in [t_0, t_{max}]\}$, is a set of endogenous variables, which are observed and every $V_t^i \in \mathbb{V}$ is functionally dependent on (directly caused by) some subset of $\mathbb{L}^{v_t^i} \cup \mathbb{V}_{\leq t} \setminus \{V_t^i\}$ where $\mathbb{V}_{\leq t} = \{V_{t'}^j \mid j \in [1, d], t' \leq t\}$. \mathbb{F} is a set of functions such that for all $V_t^i \in \mathbb{V}$, $f^{v_t^i}$ is a mapping from $\mathbb{L}^{v_t^i}$ and a subset of $\mathbb{V}_{\leq t} \setminus \{V_t^i\}$ to V_t^i . $P(\mathbb{I})$ is a joint probability distribution over \mathbb{L} .

A DTDSCM implicitly assumes that an effect cannot precede its cause. This assumption is explicitly stated as follows.

Assumption 1. Consider a DTDSCM $\mathcal{M} = (\mathbb{L}, \mathbb{V}, \mathbb{F}, P(\mathbb{I}))$. Suppose V_t^i is an endogenous variable which is functionally dependent on $\mathbb{W} \subseteq \mathbb{V} \setminus \{V_t^i\}$, i.e., $V_t^i := f^{v_t^i}(\mathbb{W}, \mathbb{L}^{v_t^i})$. For all $V_{t'}^j \in \mathbb{W} \cup \mathbb{L}^{v_t^i}$, it is assumed that $t' \leq t$.

Furthermore, stationarity is assumed.

Assumption 2. Consider a DTDSCM $\mathcal{M} = (\mathbb{L}, \mathbb{V}, \mathbb{F}, P(\mathbb{I})). \ \forall f^{v_t^i}, f^{v_{t'}^i} \in \mathbb{F}, \ f^{v_t^i} = f^{v_{t'}^i}.$

Assumption 2 have two important implications. First, it entails that if $Y_t = f^{y_t}(X_{t-1}, W_{t-1})$, then $\forall t' \in [t_0 + 1, t_{max}], Y_{t'} = f^{y_t}(X_{t'-1}, W_{t'-1})$. Furthermore, it allows us to fix the maximum temporal lag between a cause and an effect, denoted as γ_{\max} . The contributions of this paper are relevant whenever multivariate time series—where each \mathbb{V}^i such that $i \in [1, d]$ represents a time series with a single observation per temporal variable \mathbb{V}^i_t —or multivariate spatio-temporal series —where each \mathbb{V}^i such that $i \in [1, d]$ is a spatio-temporal series with multiple observations exist for each temporal variable \mathbb{V}^i_t (as seen in certain types of cohort studies)—are considered. In the former case, if Assumption 2 is violated, identifying a unique total



Figure 1: Three SCGs and six FT-ADMGs, where $\gamma_{\max} = 1$, such that FT-ADMGs 1 and 2 are compatible with SCG 1, FT-ADMGs 3 and 4 are compatible with SCG 2, and FT-ADMGs 5 and 6 are compatible with SCG 3. Each pair of red and blue vertices represents the cause and the effect of interest and green vertices are those that interpret all paths from the cause to the effect of interest. In SCG 1 and SCG 2, $\{W\}$ satisfies Definition 7 for the total effect $\Pr(y_t|do(x_{t-1}))$. However, $\{W\}$ does not satisfy Definition 7 in SCG 3 for the total effect $\Pr(y_t|do(x_{t-1}))$ since $Cycles(X, \mathcal{G}^s) \neq \emptyset$ and $\gamma \neq 0$.

effect becomes ill-posed. This is because this case assumes a dynamic system with a single multivariate observational time series, meaning that variables preceding a given variable V_t^i within the same time series \mathbb{V}^i would act as substitutes for different realizations of that variable. Violating this assumption would imply that the total effect varies over time, making estimation impossible. Furthermore, in this case, additional assumptions—such as those proposed in Eichler & Didelez (2010)—are required for the estimation of the total effect. However, in the latter case, where multiple observations of each temporal variable \mathbb{V}_t^i are available, these additional assumptions are not necessary. The distinction between these two cases is considered to be beyond the scope of this work. Therefore, I will not explicitly differentiate between them and for clarity and brevity, I will refer to the set of variables \mathbb{V}^i simply as a time series throughout the remainder of this paper.

It is also supposed that the DTDSCM can be qualitatively represented by full-time² directed acyclic graph (FT-DAG), commonly known as a full-time causal graph (Peters et al., 2013). To check the identifiability of

²The term "full-time" underscores that the graph represents the entirety of a dynamic system over time (from t_0 to t_{max}). This representation can stretch over a vast range, suggesting that any segment of a full-time DAG depicted in a figure should

a total effect, it is standard to first transform the FT-DAG to a full-time acyclic directed mixed graph (FT-ADMG) (Richardson, 2003) which is naturally derived from FT-DAGs with latent variables via an operation called latent projection (Tian & Pearl, 2002). The FT-ADMG is supposed to be a DAG with bidirected edges representing latent confounding. If all variables representing latent confounding become observed then the FT-ADMG becomes a FT-DAG. Figures 1a, 1b, 1d,1e, 1g and 1h present six different FT-ADMGs. In our setting, FT-ADMGs can also be obtained directly from the DTDSCM, as shown below.

Definition 2 (Full-Time Acyclic Directed Mixed Graph). Consider a DTDSCM \mathcal{M} . The full-time acyclic directed mixed graph (FT-ADMG) $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ induced by \mathcal{M} is defined in the following way:

$$\mathbb{E}^1 := \{ X_{t'} \to Y_t \quad |\forall Y_t \in \mathbb{V}, \ X_{t'} \in \mathbb{X} \text{ such that } Y_t := f_t^y(\mathbb{X}, \mathbb{L}^{y_t}) \text{ in } \mathcal{M} \text{ and } \mathbb{X} \subset \mathbb{V} \setminus \{Y_t\} \},\\ \mathbb{E}^2 := \{ X_{t'} \longleftrightarrow Y_t \quad |\forall X_{t'}, Y_t \in \mathbb{V} \text{ such that } \mathbb{L}^{x_{t'}} \not\perp \mathbb{L}^{y_t} \}.$$

where $\mathbb{E} = \mathbb{E}^1 \cup \mathbb{E}^2$.

FT-ADMG notions For an FT-ADMG \mathcal{G} , a *path* from $X_{t'}$ to Y_t in \mathcal{G} is a sequence of distinct vertices $\langle X_{t'}, \ldots, Y_t \rangle$ in which every pair of successive vertices is adjacent. A *directed path* from $X_{t'}$ to Y_t is a path from $X_{t'}$ to Y_t in which all edges are directed towards Y_t in \mathcal{G} , that is $X_{t'} \to \ldots \to Y_t$. A *back-door path* between $X_{t'}$ and Y_t is a path between $X_{t'}$ and Y_t with an arrowhead into $X_{t'}$ in \mathcal{G} . If $X_{t'} \to Y$, then $X_{t'}$ is a *parent* of Y_t . If there is a directed path from $X_{t'}$ to Y_t , then $X_{t'}$ is an *ancestor* of Y_t , and Y_t is a *descendant* of $X_{t'}$. A vertex counts as its own descendant and as its own ancestor. The sets of parents, ancestors and descendants of $X_{t'}$ in \mathcal{G} are denoted by $Par(X_{t'}, \mathcal{G})$, $Anc(X_{t'}, \mathcal{G})$ and $Desc(X_{t'}, \mathcal{G})$ respectively. If a path π contains $X_{t'} \to W_{t''} \leftarrow Y_t$ as a subpath, then $W_{t''}$ is a *collider* on π . A path π from $X_{t'}$ to Y_t is *active* given a vertex set \mathbb{W} , with $X_{t'}, Y_t \notin \mathbb{W}$ if every non-collider on π is not in \mathbb{W} , and every collider on π has a descendant in \mathbb{W} . Otherwise, \mathbb{W} blocks π . A set of vertices \mathbb{W} intercepts all directed paths from $X_{t'}$ to Y_t if every directed path from $X_{t'}$ to Y_t passes through at least one vertex in \mathbb{W} . Lastly, each vertex in an FT-ADMG is called a temporal vertex or a micro vertex.

The total effect (Pearl et al., 2000) between two micro variables is written as $P(Y_t = y_t | do(X_{t-\gamma} = x_{t-\gamma}))$. Y_t corresponds to the response and $do(X_{t-\gamma} = x_{t-\gamma})$ represents an intervention (as defined in Pearl et al. (2000) and Eichler & Didelez (2007, Assumption 2.3)) on the variable X at time $t - \gamma$, with $\gamma \ge 0$. Unlike γ_{max} (which represents the maximum possible lag between a cause and effect), γ simply specifies the lag of interest for the query posed by the user. In the remainder of the paper, γ is considered to be in $\{0, \gamma_{max}\}$, and, with a slight abuse of notation, $P(Y_t = y_t \mid do(X_{t-\gamma} = x_{t-\gamma}))$ is written as $P(y_t \mid do(x_{t-\gamma}))$. The identifiability of the total effect in FT-ADMGs is defined as follows.

Definition 3 (Identifiability of total effects in FT-ADMGs). Let $X_{t-\gamma}$ and Y_t be distinct vertices in an FT-ADMG $\mathcal{G} = (\mathbb{V}, \mathbb{E})$. The total effect of $X_{t-\gamma}$ on Y_t is identifiable in \mathcal{G} if $\Pr(y_t|do(x_{t-\gamma}))$ is uniquely computable from any positive observational distribution consistent with \mathcal{G} .

A total effect is uniquely computable if $\Pr(y_t \mid do(x_{t-\gamma}))$ can be expressed using a *do-free formula*. Given a fully specified FT-ADMG, there exists many tools to identify the total effect. For example the standard backoor criterion (Pearl, 1993b; 1995) can be used to find a set of covariates \mathbb{B} that is sufficient for adjustment; in such case the do-free formula of the total effect is written as $\sum_{\mathbb{D}} \Pr(y_t \mid x_{t-\gamma}, \mathbb{D}) \Pr(\mathbb{D})$. When such a set does not exist due to latent confounding, the standard front-door criterion, as introduced by Pearl (1995), can sometimes enable the derivation of an alternative do-free formula.

However, in many real-world applications such as medicine or epidemiology, experts often cannot provide the FT-ADMG. Furthermore, discovering the true FT-ADMG from real observational data is challenging (Spirtes et al., 2000; Mogensen et al., 2018; Runge et al., 2019; Assaad et al., 2022) because 1) causal discovery algorithms rely on untestable assumptions that are not always satisfied in real applications (Aït-Bachir et al., 2023) and 2) even when the assumptions hold, the output of such methods does not always correspond to the true graph but rather to a class of graphs that include the true graph (Gerhardus & Runge, 2020). Therefore,

be viewed as merely a snapshot of the larger graph. Under Assumption 2, it is sometimes possible to extrapolate the entire graph from such a snapshot.

experts typically rely on a partially specified representation of the FT-ADMG, known as a summary causal graph³.

Definition 4 (Summary Causal Graph with possible latent confounding). Consider an FT-ADMG $\mathcal{G} = (\mathbb{V}, \mathbb{E})$. The summary causal graph (SCG) $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ compatible with \mathcal{G} is defined in the following way:

$$\begin{split} \mathbb{S} &:= \{ V^i = (V^i_{t_0}, \cdots, V^i_{t_{max}}) & | \forall i \in [1, d] \}, \\ \mathbb{E}^{s1} &:= \{ X \to Y & | \forall X, Y \in \mathbb{S}, \ \exists t' \leq t \in [t_0, t_{max}] \text{ such that } X_{t'} \to Y_t \in \mathbb{E} \}, \\ \mathbb{E}^{s2} &:= \{ X \leftarrow \cdots \to Y & | \forall X, Y \in \mathbb{S}, \ \exists t', t \in [t_0, t_{max}] \text{ such that } X_{t'} \leftarrow \cdots \to Y_t \in \mathbb{E} \}. \end{split}$$

where $\mathbb{E}^s = \mathbb{E}^{s1} \cup \mathbb{E}^{s2}$.

SCG notations For an SCG \mathcal{G}^s , a directed path from X to Y and the edge $Y \to X$ form a *directed cycle* in \mathcal{G}^s . Additionally, the specific case where a vertex has a directed edge to itself (self loop) is considered as a cycle too. $Cycles(X, \mathcal{G}^s)$ denotes the set of all directed cycles containing X in \mathcal{G}^s . A *directed path* between X and Y is a path between X and Y which starts by $X \to$ and does not contain any arrow on the path pointing strictly towards X. A *back-door path* between X and Y is a path between X and Y which starts by $X \to$ and does not contain any arrow on the path pointing strictly towards X. A *back-door path* between X and Y is a path between X and Y which starts by either $X \leftarrow$ or $X \leftarrow \cdots$ or $X \rightleftharpoons$. If $X \to Y$ or $X \leftrightarrows Y$, then X is a *parent* of Y. The notions of ancestors and descendants are defined similarly as in the case of FT-ADMGs. A path in an SCG is blocked given a set \mathbb{W} if it contains a strict collider at W (i.e., $\to W \leftarrow$, and not $\leftrightarrows W \leftarrow$ or $\to W \leftrightarrows$) such that $\mathbb{W} \cap Desc(W, \mathcal{G}^s) = \emptyset$ or if it contains strict non-collider at W (i.e., $\to W \to$ or $\leftarrow W \to$ or $\leftarrow W \leftrightarrows$, and not $\leftrightarrows W \leftarrow$) such that $W \in \mathbb{W}$ and there exists a directed edge pointing from W to a vertex on the path that does not form a cycle with W. A path in an SCG is activated if it is not blocked. In particular, a path is activated by an empty set if it does not contain any strict collider. The notions of interception is defined similarly as in the case of FT-ADMGs. Lastly, each vertex in an SCG is called a cluster or a macro vertex and it represents a time series.

Many FT-ADMGs might share the same compatible SCG. For example, Figure 1c presents the SCG compatible with the two FT-ADMGs in Figures 1a and 1b, Figure 1f presents the SCG compatible with the two FT-ADMGs in Figures 1d and 1e, and Figure 1i presents the SCG compatible with the two FT-ADMGs in Figures 1g and 1h. For a given SCG \mathcal{G}^s , any FT-ADMG from which \mathcal{G}^s can be derived is called as a *candidate FT-ADMG* for \mathcal{G}^s . The set of all candidate FT-ADMGs for \mathcal{G}^s is denoted by $\mathcal{C}(\mathcal{G}^s)$.

This paper focues on identifying the total effect when the only knowledge one has of the underlying DTDSCM consists in the SCG derived from the unknown, true FT-ADMG. In this setting, the identifiability of the total effect in SCGs is defined as follows:

Definition 5 (Identifiability of total effects in SCGs). Consider an SCG \mathcal{G}^s . Let $X_{t-\gamma}$ and Y_t be distinct vertices in every candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$. The total effect of $X_{t-\gamma}$ on Y_t is identifiable in \mathcal{G}^s if $\Pr(y_t|do(x_{t-\gamma}))$ is uniquely computable from any positive observational distribution consistent with any FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Obviously, when the true FT-ADMG is unknown but the compatible SCG is accessible, it is possible to enumurate all candidates FT-ADMGs and then search for a do-free formula applicable for each of those FT-ADMGs. Within this approach, it is possible to identify the total effect if it is possible to find a set of micro vertices that satisfies the front-door criterion for each FT-ADMG in $C(\mathcal{G}^s)$. However, enumerating all candidate FT-ADMGs is computationally expensive (Robinson, 1977), even when considering the constraints given by an SCG. Therefore, this paper addresses the following technical problem:

³One key motivation for employing summary causal graphs stems from the current limitations of causal discovery methods, which often struggle in practical applications due to their reliance on non-testable strong assumptions. Particularly in fields like medicine and epidemiology, researchers tend to prefer graphs built from prior knowledge rather than those inferred purely from data. However, fully specified graphs are very complicated to construct and validate manually and that is why it is important to work with (and ask experts to build) partially specified graphs such as summary causal graphs. That being said, recent studies have demonstrated that inferring summary causal graphs from data is more feasible than inferring FT-ADMGs from data (Wahl et al., 2024). This supports the relevance of our work even if researchers choose to utilize causal discovery, suggesting that our approach remains applicable when taking a data-driven approach to get the summary causal graph.

Problem 1. Consider an SCG \mathcal{G}^s and the total effect $\Pr(y_t|do(x_{t-\gamma}))$. The aim is to find sufficient conditions for identifying $\Pr(y_t|do(x_{t-\gamma}))$ using an SCG with latent confounding without enumerating all candidate FT-ADMGs in $\mathcal{C}(\mathcal{G}^s)$.

3 Unsuitability of the standard front-door criterion when applied to SCGs

This section elucidates why the standard front-door criterion does not straightforwardly apply to SCGs. The standard front-door criterion consists of three conditions. Initially, the criterion was introduced for ADMGs (which means it can also be correctly applied to an FT-ADMG), but in order to illustrate its unsuitability in the context of this paper, it is presented here with few modification (in blue) given Definition 6.

Definition 6 (Standard front-door criterion naively applied to SCGs). Consider an SCG \mathcal{G}^s . A set of macro vertices \mathbb{W} in \mathcal{G}^s satisfy the front-door criterion relative to a pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with a pair of macro vertices (X, Y) in \mathcal{G}^s if:

- 1. W intercepts all activated directed paths from X to Y;
- 2. there is no activated back-door path from X to \mathbb{W} ;
- 3. all back-door paths from \mathbb{W} to Y are blocked by X;

To obtain the standard front-door criterion for FT-ADMGs, the terms in blue need to be modified as follows: since the variables of interest $(X_{t-\gamma}, Y_t)$ are already vertices in the given graph (FT-ADMG), there's no need to map these vertices to their compatible counterparts in the SCG. Therefore, the phrase "compatible with a pair of macro vertices (X, Y) in \mathcal{G}^{s} should be removed. Next, replace all remaining instances of "SCGs" with "FT-ADMGs" and " \mathcal{G}^s " with " \mathcal{G} ". Additionally, since FT-ADMGs do not include macro vertices, the term "macro" should be replaced with "micro". Finally, replace all remaining instances of "X" and "Y" with $X_{t-\gamma}$ and Y_t . Pearl's insight is that if there exists a set \mathbb{W} that intercepts all directed paths from $X_{t-\gamma}$ to Y_t and there is no hidden confounding that cannot be blocked by $X_{t-\gamma}$, the total effect of $X_{t-\gamma}$ on Y_t can be identified. This is achieved by: (i) identifying the effect of $X_{t-\gamma}$ on \mathbb{W} (which is identifiable because the unobserved confounders influence $X_{t-\gamma}$ but not \mathbb{W} ; (ii) identifying the effect of \mathbb{W} on Y_t conditional on $X_{t-\gamma}$ (which is identifiable because the unobserved confounders affect Y_t but not \mathbb{W}); and (iii) multiplying the do-free formulas $\Pr(x_{t-\gamma} \mid w)$ and $\Pr(y_t \mid x_{t-\gamma}, w) \Pr(x_{t-\gamma})$. Intuitively, in the context of an FT-ADMG (or an ADMG), the standard front-door criterion enables the identification of a total effect that cannot be identified through adjustment alone, by decomposing it into two identifiable total effects. The previously proposed extension of the front-door criterion to time series settings Eichler & Didelez (2010)-referred to here as the Granger front-door criterion-equivalent to the one given in Definition 6 with one minor change which makes slightly more restrictive: the Granger front-door criterion forbid all back-door paths from Xto \mathbb{W} (activated and not activated) in Condition 2. In contrast to the criterion proposed in the next section, the Granger front-door criterion relies on two key assumptions: it implicitly assumes there is no latent confounding across different time points within a single time series, and it explicitly rules out instantaneous causal relationships. Under these assumptions-i.e., when the SCG reflects the absence of latent confounding over time and no instantaneous effects-the Granger front-door criterion allows identification of a total effect that is not identifiable via standard adjustment, by decomposing it into two identifiable total effects.

It turns out that directly applying the standard front-door criterion as introduced in Pearl (1995) to SCGs is not suitable. The main reason for this unsuitability is that not having a back-door path between two macro vertices in the SCG does not imply that there is no back-door path between two compatible micro vertice in any FT-ADMG in $C(\mathcal{G}^s)$. Which means that even if Conditions 2 and 3 of Definition 6 are satisfied in the SCG \mathcal{G}^s , this does not guarantee that Conditions 2 and 3 of the standard front-door criterion are met for any FT-ADMG within $C(\mathcal{G}^s)$. Sometimes, this would imply that even if the standard front-door criterion is satisfied when applied to SCGs, the total effect of interest might be non identifiable. For illustratation consider the total effect $P(y_t \mid x_{t-1})$ and the SCG in Figure 1i where W satisfies the standard front-door criterion with respect to X and Y. However, the total effect is not identifiable using W because in the FT-ADMG in Figure 1g or Figure 1h (which is assumed to be unknown), there is a back-door path between X_{t-1} and W_t passing by X_t which should not be blocked by X_t (since X_t is a descendant of X_{t-1} and ancestor of W_t). Similarly, the Granger front-door criterion would also incorrectly conclude that the total effect is identifiable. Recall that the Granger front-door criterion assumes the absence of latent confounding across different time points within a single time series, and explicitly assumes no instantaneous causal relationships and both of these assumptions are violated in the ADMGs shown in Figure1g and Figure 1h. Therefore, it is not surprising that, when relying solely on the SCG and lacking information about the corresponding FT-ADMG, the Granger front-door criterion would mistakenly suggest identifiability of the total effect.

Let us also consider another similar example, with a similar SCG but where $X \to X$ is not in the SCG but there is a larger cycle containing X, like the SCG in Figure 3b and suppose that our aim is to identify the total effect $\Pr(y_t \mid do(X_{t-1}))$ with $\gamma_{\max} = 1$. It is possible to construct an FT-ADMG compatible with this SCG in which a back-door path exists: $X_{t-1} \leftarrow U_{t-1} \to X_t \to W_t$. This path cannot be blocked by conditioning on X_t because the presence of a self loop on X implies that X_t could be a descendant of X_{t-1} . At the same time, blocking this path via U_{t-1} is also problematic, because it is possible to imagine an alternative FT-ADMG, also compatible with the given SCG, where the same path is instead a directed path: $X_{t-1} \to U_{t-1} \to X_t \to W_t$, implying that U_{t-1} is a descendant of X_{t-1} . In this case, conditioning on U_{t-1} would block a valid directed path, further complicating identification. However, in such cases, applying the standard front-door criterion or the Granger front-door criterion would incorrectly suggest that the total effect is identifiable—when, in reality, it is not. Again, this is unsurprising, as these tools were not designed for the specific setting considered in this paper.

At an intuitive level, the failure of the standard front-door criterion and of the Granger front-door criterion in these cases arises due to the simultaneous presence of instantaneous relations between time series and either $X \leftrightarrow X$ and a self loop on X in \mathcal{G}^s or just simply larger cycles on X. The existence of $X \leftarrow X$ in \mathcal{G}^s implies that there may be a back-door path between $X_{t-\gamma}$ (for $\gamma > 0$) and W_t , starting with $X_{t-\gamma} \leftarrow X_t$. Meanwhile, the presence of a self loop on X means that this back-door path cannot be blocked using X_t , since X_t would be a descendant of $X_{t-\gamma}$. As a result, conditioning on X_t would block a directed path from $X_{t-\gamma}$ to W_t , leading to bias in the estimation of the total effect of $X_{t-\gamma}$ to W_t . It is important to note that this kind of back-door paths is not immediately apparent when examining only the SCG with classical tools built for DAGs or ADMGs such as the standard front-door criterion. The same intuition holds for the case when there is larger cycles containing X even in the absence of $X \leftarrow X$ in the SCG.

In other cases, the unsuitability of the standard front-door criterion when applied to SCGs might imply that, even if the total effect of interest is identifiable, the corresponding do-free formula could be more complex than the one associated with the standard front-door criterion (Pearl, 1995) or the Granger frontdoor criterion (Eichler & Didelez, 2010). For example, consider the total effect $P(y_t \mid x_{t-1})$ and the SCG in Figure 1c (no cycle containing X in the SCG but $X \leftarrow X$ is in the SCG), where W satisfies the standard front-door criterion with respect to X and Y. Now, consider one of the FT-ADMGs given in Figures 1a and 1b compatible with this SCG. In this FT-ADMG, the micro vertices (i.e., $\{W_{t-1}, W_t\}$) corresponding to the macro vertex W, which intercepts all directed paths from X_{t-1} to Y_t , do not satisfy the standard front-door criterion for (X_{t-1}, Y_t) , since there exists a back-door path $X_{t-1} \leftrightarrow X_t \rightarrow W_t$. However, since there is no cycle containing X in the SCG, this back-door path can be easily blocked by adjusting for X_t . Doing so results in a do-free formula similar to the one derived using the front-door criterion, but with the additional step of adjusting for X_t . These types of do-free formulas are derived by combining the backdoor criterion with the front-door criterion, as discussed in (Pearl, 1995) and subsequently applied by other authors (Fulcher et al., 2019). However, it is important to note that the same do-free formula should not be used if the SCG in Figure 1f (there is cycle containing X in the SCG but $X \leftrightarrow X$ is not in the SCG) is considered. Moreover, distinguishing between these different scenarios is not feasible when relying solely on the SCG, whether using classical tools developed for DAGs or ADMGs, such as the standard front-door criterion, or tools tailored to SCGs that assume no latent confounding across time and no instantaneous relations, such as the Granger front-door criterion. These types of unsuitability also exists when considering instantaneous total effects. For instance, take the SCG in Figure 2b and the total effect of X_t on Y_t . In this case, there are no back-door paths between X and W within the SCG. However, it is possible to construct an FT-ADMG where a back-door path exists: $X_t \leftarrow U_t \leftarrow \ldots \leftarrow U_{t-1000} \rightarrow X_{t-1000} \rightarrow W_{t-1000} \rightarrow \ldots \rightarrow W_t$ (assuming $t_0 < t - 1000$). This path must be blocked to ensure correct identification using the front-door formula. To block this and similar paths, one would need to either adjust on all variables between X_{t_0} and X_{t-1} (which is impractical) or simply adjust on $U_t, U_{t-1}, \ldots, U_{t-\gamma_{\max}}$. However, the standard front-door criterion and its associated theorem cannot be applied to infer this requirement.

4 The identifiability of total effects in SCGs with latent confounding

This section, presents the main results of the paper, with some of the corresponding proofs omitted and provided in the appendix. Let us start by giving the extension of the front-door criterion for SCGs.

Definition 7 (SCG-front-door criterion). Consider an SCG \mathcal{G}^s . A set of macro vertices \mathbb{W} in \mathcal{G}^s satisfy the SCG-front-door criterion relative to a pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with a pair of macro vertices (X, Y) in \mathcal{G}^s if:

- 1. \mathbb{W} intercepts all activated directed paths from X to Y;
- 2. there is no activated back-door path from X to W;
- 3. all back-door paths from \mathbb{W} to Y are blocked by X;
- 4. one of the following holds:
 - (a) $Cycles(X, \mathcal{G}^s) = \emptyset$; or
 - (b) $Cycles(X, \mathcal{G}^s) = \{X \to X\}$ and $\exists Z \in Anc(X, \mathcal{G}^s)$ such that $X \leftarrow Z$ in \mathcal{G}^s ; or
 - (c) $\gamma = 0$.

Conditions 1-3 in Definition 7 correspond to the three conditions in the standard front-door criterion (Pearl, 1995). For an illustration of Definition 7, Figure 2 provides several examples of SCGs, including the SCGs given in Figures 1f and Figures 1i, where Definition 7 is satisfied for W relative to $(X_{t-\gamma}, Y_t)$. Notice that Definition 7 remains satisfied for W relative to (X_t, Y_t) if a cycle is added on X that does not involve any other vertex in the presented SCGs, as illustrated in Figure 3 which includes the SCG given in Figure 1c. Note that in these SCGs, W is not necessarily the only vertex satisfying Definition 7, for example, in Figure 2e, U, also satisfied the SCG-front-door criterion. In contrast, Figure 4 provides several examples of SCGs where Definition 7 is not satisfied for any vertex relative to $(X_{t-\gamma}, Y_t)$.

In the following, lemmas that form the building blocks for the theorem introduced at the end of the section is presented. The first lemma asserts that if a set of macro vertices intercepts all directed paths between X and Y in an SCG, then there exists a finite set of micro vertices that intercepts all directed paths between $X_{t-\gamma}$ and Y_t in any candidate FT-ADMG. This mirrors the first Condition of the standard front-door criterion for ADMGs.

Lemma 4.1. Consider an SCG \mathcal{G}^s . If a set of macro vertices \mathbb{W} intercepts all directed paths from X to Y in \mathcal{G}^s then $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}$ intercepts all directed paths from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

For an illustration of this lemma, consider any FT-ADMG depicted in Figure 1 or Figure 5, where in the corresponding SCG, all directed paths from X to Y are intercepted by W. Observe that any directed path from X_{t-1} to Y_t must pass through a micro vertex $W_{t-\lambda}$ that corresponds to W. If $\lambda > \gamma$, then all active paths from X_{t-1} to Y_t passing through $W_{t-\lambda}$ are not directed paths since $W_{t-\lambda}$ is temporally prior to t and $t - \gamma$. Conversely, if $\lambda < 0$, all paths from X_{t-1} to Y_t that pass through $W_{t-\lambda}$ are blocked, as $W_{t-\lambda}$ would always act as a collider due to its temporal positioning after both t and $t - \gamma$.

The second lemma asserts that given Conditions 1-3, along with Condition 4a, it is guaranteed that there exists a finite set that blocks all back-door paths from $X_{t-\gamma}$ to any vertex in any set of micro vertices that intercepts all directed paths from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG. Furthermore, this finite set does not contain any descendants of $X_{t-\gamma}$. This mirrors the second Condition of the standard front-door criterion for ADMGs.

Lemma 4.2. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. If $Cycles(X, \mathcal{G}^s) = \emptyset$ then for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \le \ell \le \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{0 \le \ell \le \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(B_{t-\gamma-\ell})_{1 \le \ell \le \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma+\ell})_{1 \le \ell \le \gamma}\}$ blocks all back-door paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ and does not contain any descendant of $X_{t-\gamma}$ in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

The intuition behind Lemma 4.2 is that conditioning on micro variables corresponding to the parents of X in the SCG blocks all back-door paths that start with $X_{t-\gamma} \leftarrow$. This still leaves some possible back-door paths



Figure 2: Ten SCGs satisfying Definition 7 for W relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$, $\forall \gamma \in \{0, \dots, \gamma_{\max}\}$, i.e., for the first half of these SCGs, Conditions 1, Conditions 2, Conditions 3, and Conditions 4a in Definition 7 are satisfied and for the second half, Conditions 1, Conditions 2, Conditions 3, and Conditions 4b in Definition 7 are satisfied. Each pair of red and blue vertices represents the total effect of interest and green vertices are those that intercept all directed paths from the cause to the effect of interest.

active, those starting with $X_{t-\gamma} \longleftrightarrow$, which require additional conditioning sets to be blocked, namely some micro variables corresponding to the ancestors of \mathbb{W} . The set $\{X_{t-2}, X_t\}$ satisfies the conditions of this lemma when considering the total effect of X_{t-1} on Y_t and the SCG presented in Figure 1c. Specifically, $\{X_{t-2}, X_t\}$ effectively blocks all back-door paths from X_{t-1} to $\{W_{t-1}, W_t\}$ and does not include any descendant of X_{t-1} in any FT-ADMG that is compatible with the SCG (e.g., Figure 1a, Figure 1b). For another, more complete, visual explanation of why the selected set in Lemma 4.2 blocks all back-door paths, refer to the FT-ADMG in Figure 5a. In the figure, the vertices corresponding to $\{(B_{t-\gamma-\ell})_{0\leq\ell\leq\gamma_{\max}}|B\in Par(X,\mathcal{G}^s)\}$ are highlighted in brown, the vertices corresponding to $\{(B_{t-\gamma-\ell})_{1\leq\ell\leq\gamma_{\max}}|B\in Anc(\mathbb{W},\mathcal{G}^s)\cap Desc(X,\mathcal{G}^s)\}$ are highlighted in pink, and the vertices corresponding to $\{(X_{t-\gamma+\ell})_{1\leq\ell\leq\gamma}\}$ are highlighted in gray.

Notice that if $Cycle(X, \mathcal{G}^s) \neq \emptyset$, then there could exist an FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$ where micro vertices related to X that temporally succeed $X_{t-\gamma}$ can be descendants of $X_{t-\gamma}$. Additionally, these descendants can share a latent confounder with $X_{t-\gamma}$, and the path responsible of this latent confounding cannot be blocked. For example, in the FT-ADMG in Figure 1g compatible with the SCG in Figure 1i that contain a cycle of size 1 on X, it is clear that X_t is a descendant of X_{t-1} and that the back-door path from X_{t-1} to W_t starting with $X_{t-1} \leftrightarrow X_t$ cannot be blocked by any vertex. Thus, $\{W_{t-1}, W_t\}$ cannot be used to intercept the relationship between X_{t-1} and Y_t for identification using the front-door criterion. However, it turns out this



Figure 3: Five SCGs satisfying Definition 7 for W relative only to the pair of micro vertices (X_t, Y_t) , i.e., Conditions 1, Conditions 2, Conditions 3, and Conditions 4c in Definition 7 are satisfied. Each pair of red and blue vertices represents the total effect of interest and green vertices are those that intercept all directed paths from the cause to the effect of interest.

is not the case when either the only cycle on X is a self loop and there is no latent confounding between X and its ancestors—which implies that $X \leftrightarrow X$ is not in \mathcal{G}^s (as in the SCG in Figure 1f)—or when $\gamma = 0$, as stated in the following two lemmas.

Lemma 4.3. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. If $Cycle(X, \mathcal{G}^s) = \{X \to X\}$ and $|\exists Z \in Anc(X, \mathcal{G}^s)$ such that $X \leftrightarrow Z$ in \mathcal{G}^s then for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{\max}}\}$ blocks all back-door paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ and does not contain any descendant of $X_{t-\gamma}$ in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

The intuition behind Lemma 4.3 is that, similarly to the previous lemma, conditioning on micro variables (for $t-\gamma$ to $t-\gamma-\gamma_{max}$) corresponding to the parents of X in the SCG blocks all back-door paths that start with $X_{t-\gamma} \leftarrow U_{t-\lambda'}$ where $X \neq U$. This leaves only possible back-door paths starting with $X_{t-\gamma} \leftarrow X_{t-\lambda'}$, which can be blocked by timepoints of X that preceeds $t-\gamma$. The set $\{U_{t-2}, U_{t-1}, X_{t-2}\}$ statisfies Lemma 4.3 for the total effect of X_{t-1} on Y_t when considering the SCG in Figure 1f. Which means the set $\{U_{t-2}, U_{t-1}, X_{t-2}\}$ blocks all the back-door paths between X_{t-1} and W_t in any FT-ADMG compatible with the SCG considered. To visually see why the selected set in Lemma 4.3 blocks all back-door paths, refer to the FT-ADMG in Figure 5b. In the figure, the vertices corresponding to $\{(B_{t-\ell})_{0 \leq \ell \leq \gamma_{max}} | B \in Par(X, \mathcal{G}^s)\}$ are highlighted in brown and the vertices corresponding to $\{(X_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{max}}\}$ are highlighted in pink.

Compared to Lemma 4.2, in Lemma 4.3, the set $(X_{t-\gamma+\ell}) 1 \le \ell \le \gamma$ is not included in the conditioning set for two main reasons: (1) in the context of Lemma 4.3, where self loops are allowed, the temporal variables in this set may be descendants of $Xt - \gamma$, and (2) in this scenario, no back-door paths start with $\leftarrow \rightarrow \rightarrow$. For the same reason as (2), the only subset considered from $(B_{t-\gamma-\ell}) 1 \le \ell \le \gamma \max | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G}^s)$ is $(X_{t-\gamma-\ell}) 1 \le \ell \le \gamma \max$.

Remark that Lemma 4.3 would fail if larger cycles involving X were considered, meaning cycles on X beyond a simple self loop as discussed in Section 3.

Let us now turn to a lemma, which permits even larger cycles and allows confounding between X and its ancestors, but imposes the restriction that γ must be zero.

Lemma 4.4. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices (X_t, Y_t) compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices



Figure 4: Five SCGs not satisfying Definition 7 for any vertex relative to the micro of vertices $(X_{t-\gamma}, Y_t)$. The SCG in (a) and (d) does not satisfy Definition 7 because Condition 2 is not satisfied. The SCG in (b) and (e) does not satisfy Definition 7 because Condition 3 is not satisfied. The SCG in (c) does not satisfy Definition 7 because Condition 2 and 3 are not satisfied. Each pair of red and blue vertices represents the total effect of interest and green vertices are those that intercepts all directed paths from the cause to the effect of interest.

 $\begin{array}{ll} (X_t,Y_t). & Then \ for \ any \ W_t \in \mathbb{W}_t, \ the \ set \ \{(B_{t-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Anc(X,\mathcal{G}^s) \setminus Desc(X,\mathcal{G})\} \cup \\ \{(B_{t-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in (Anc(X,\mathcal{G}^s) \cup Anc(\mathbb{W},\mathcal{G}^s)) \cap Desc(X,\mathcal{G}^s)\} \ blocks \ all \ back-door \ paths \ from \ X_t \ to \ W_t \ and \ does \ not \ contain \ any \ descendant \ of \ X_t \ in \ any \ candidate \ FT-ADMG \ in \ C(\mathcal{G}^s), \end{array}$

The intuition behind Lemma 4.4 is that conditioning on micro variables corresponding to the parents of X in the SCG cannot block all back-door paths because a parent can also be a descendant of X (since cycles are allowed in this case), but the set micro variables (for t to $t - \gamma_{max}$) corresponding to the ancestors of X that are not descendants of X in the SCG and the set micro variable (for t - 1 to $t - \gamma_{max}$) (a subset of the set in purple) corresponding to the ancestors of X that are descendants of X in the SCG blocks all back-door paths that start with \rightarrow . This still leaves some possible back-door paths active, those starting with $\leftarrow \cdots$, which require additional conditioning sets to be blocked, namely some of the ancestors of W. The set $\{X_{t-1}, W_{t-1}\}$ statisfies Lemma 4.4 for the total effect of X_t on Y_t when considering any of the SCGs given in Figures 1c, 1f and 1i. Which means the set $\{X\}$ blocks all the back-door paths between the X_t and W_t in any FT-ADMG compatible with the SCG considered. To visually see, on a more complete example, why the selected set in Lemma 4.4 blocks all back-door paths, refer to the FT-ADMG in Figure 5c. In the figure, the vertices corresponding to $\{(B_{t-\ell})_{0\leq \ell\leq \gamma_{max}} | B \in Anc(X, \mathcal{G}^s) \cup Anc(W, \mathcal{G}^s)) \cap Desc(X, \mathcal{G})\}$ are highlighted in pink.

Notice that, compared to Lemma 4.2, in Lemmas 4.3 and 4.4, two key replacements are made regarding the selected conditioning sets: $\{(B_{t-\ell})_{0 \le \ell \le \gamma \max} | B \in Par(X, \mathcal{G}^s)\}$ is replaced with $\{(B_{t-\ell})_{0 \le \ell \le \gamma \max} | B \in Anc(X, \mathcal{G}^s) \setminus Desc(X, \mathcal{G})\}$ and $\{(B_{t-\gamma-\ell})_{1 \le \ell \le \gamma \max} | B \in Anc(W, \mathcal{G}^s) \cap Desc(X, \mathcal{G}^s)\}$ is replaced with $\{(B_{t-\ell})_{1 \le \ell \le \gamma \max} | B \in (Anc(X, \mathcal{G}^s) \cup Anc(W, \mathcal{G}^s)) \cap Desc(X, \mathcal{G})\}$ to account for potential cycles involving X. For example, in the FT-ADMG shown in Figure 5c, conditioning only on the parents, specifically U_t and U_{t-1} (without conditioning on Z_t and Z_{t-1}), would be sufficient to block all back-door paths. However, there exists another FT-ADMG compatible with the SCG in Figure 3b, which is almost identical to the FT-ADMG in Figure 5c except that now $\forall t \in [t_0, t_{max}], X_t \to U_t$ (which means $U_t \not\rightarrow X_t$). In this FT-ADMG, U_t becomes a collider between X_t and Z_t , so conditioning on it would activate a new path that cannot be blocked unless the ancestors of X_t are also included in the conditioning set.

The last lemma asserts that given Conditions 1-3, it is guaranteed that there exists a finite set that blocks all back-door paths from some micro vertices (that intercept all directed paths from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG) to Y_t . Furthermore, this finite set does not contain any descendants of the micro vertices that intercept all directed paths. This mirrors the third Condition of the standard front-door criterion for ADMGs.

Lemma 4.5. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. Then, for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{-\gamma \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \setminus Desc(\mathbb{W}, \mathcal{G}^s)\} \cup \{(B_{t-\gamma+\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(\mathbb{W}, \mathcal{G}^s)\}$ blocks all back-door paths from $W_{t-\lambda}$ to Y_t not passing by $\{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\} \setminus \{W_{t-\lambda}\}$ and it does not contain any descendent of $W_{t-\lambda}$.

The intuition behind Lemma 4.5 is much simpler than the intuition behind the previous lemmas since according to the conditions of Definition 7, there cannot be any back-door paths from $W_{t-\lambda}$ to Y_t starting with \leftarrow and cycles on W would not create any issues since they either implies active back-door paths that pass by $W_{t-\lambda'}$ such that $\lambda' > \gamma$ which can be easily blocked a set of micro variables, conceptually very similar to the one used in Lemma 4.3 and 4.4. Or they pass by $W_{t-\lambda'}$ such that $W_{t-\lambda'} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\} \setminus \{W_{t-\lambda}\}$. The set $\{W_{t-2}, X_{t-2}, X_{t-1}, X_t\}$ satisfies the conditions of this lemma when analyzing the total effect of X_{t-1} on Y_t in any of the SCGs presented in Figures 1c, 1f, and 1i. Specifically, $\{W_{t-2}, X_{t-2}, X_{t-1}, X_t\}$ effectively blocks all back-door paths from W_t (respectively, W_{t-1}) to Y_t that do not pass through W_{t-1} (respectively, W_t). Additionally, this set does not contain any descendant of W_t or W_{t-1} in any FT-ADMG that is compatible with any of these SCGs (e.g., Figures 1a, 1b, 1d, 1e, 1g, or 1h). To gain a clearer visual understanding, using a more complete example, why the selected set in Lemma 4.5 blocks all backdoor paths not passing by $\{(W_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}\setminus\{W_{t-\lambda}\}$, refer to the FT-ADMs in Figure 5d. In these FT-ADMGs, the vertices corresponding to $\{(B_{t-\gamma-\ell})_{-\gamma \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \setminus Desc(\mathbb{W}, \mathcal{G}^s)\}$ are highlighted in orange, the vertices corresponding to $\{(B_{t-\gamma-\ell})_{1 \le \ell \le \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(\mathbb{W}, \mathcal{G}^s)\}$ are highlighted in purple, and the vertices corresponding to $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}$ are highlighted in green. The vertex $X_{t-\gamma}$ is highlighted in red and in orange since it is the main cause of interest and at the same time it is in the set $\{(B_{t-\gamma-\ell})_{-\gamma\leq\ell\leq\gamma_{\max}}|B\in Anc(W,\mathcal{G}^s)\setminus Desc(W,\mathcal{G}^s)\}.$

In the following, a sound theorem is introduced for identifying total effects using SCG, even in the presence of cycles and latent confounders.

Theorem 1. If a set of vertices \mathbb{W} satisfies the SCG-front-door criterion (i.e., Definition 7) relative to $(X_{t-\gamma}, Y_t)$ and if $\Pr(\mathfrak{s}) > 0$ then $\Pr(y_t \mid do(x_{t-\gamma}))$ is identifiable and is given by the do-free formula:

$$\Pr(y_t \mid do(x_{t-\gamma})) = \sum_{\mathbb{f}} \sum_{\mathbb{b}^x} \Pr(\mathbb{f} \mid x_{t-\gamma}, \mathbb{b}^x) \Pr(\mathbb{b}^x)$$

$$\times \sum_{\mathbb{b}^f, x'_{t-\gamma}} \Pr(y_t \mid \mathbb{f}, \mathbb{b}^f, x'_{t-\gamma}) \Pr(\mathbb{b}^f, x'_{t-\gamma})$$
(1)

where $\mathbb{F} = \{ (\mathbb{W}_{t-\gamma+\ell})_{0 \le \ell \le \gamma} \};$

$$\begin{split} \mathbb{B}^{x} = & \{ (B_{t-\gamma-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Anc(X, \mathcal{G}^{s}) \setminus Desc(X, \mathcal{G}) \} \\ \cup \{ (B_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in (Anc(X, \mathcal{G}^{s}) \cup Anc(\mathbb{W}, \mathcal{G}^{s})) \cap Desc(X, \mathcal{G}) \} \cup \mathbb{B}^{c} \\ such that \ \mathbb{B}^{c} = \{ (X_{t-\gamma+\ell})_{1 < \ell < \gamma} \} if Condition \ 4a \ is \ satisfied, \ and \ \mathbb{B}^{c} = \emptyset \ otherwise; \end{split}$$

and

$$\mathbb{B}^{f} = \{ (B_{t-\gamma-\ell})_{-\gamma \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^{s}) \setminus Desc(\mathbb{W}, \mathcal{G}^{s})) \} \\ \cup \{ (B_{t-\gamma-\ell})_{1 < \ell < \gamma_{\max}} | B \in (Anc(\mathbb{W}, \mathcal{G}^{s}) \cap Desc(\mathbb{W}, \mathcal{G}^{s})) \} \setminus \{ X_{t-\gamma} \}.$$

Proof. By Lemma 4.1, since \mathbb{V} intercepts all directed paths from X to Y in the SCG, then $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}$, i.e., \mathbb{F} , intercepts all directed paths from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$. Given this result, by the law of total probability, the total effect $\Pr(y_t \mid do(x_{t-\gamma}))$ can be computed in two steps: first, by computing $\Pr(\mathbb{F} \mid do(x_{t-\gamma}))$ and $\Pr(y_t \mid do(\mathbb{F}))$, and subsequently multiplying the two quantities together while summing over \mathbb{F} .

Notice that the set used in Lemma 4.2 and the set used in Lemma 4.3 is a subset of \mathbb{B}^x and at the same time \mathbb{B}^x cannot contain any descendant of $X_{t-\gamma}$ in the context of Lemma 4.2 and in the context of Lemma 4.3.



(a) A candidate FT-ADMG of the SCG in Figure 2b showing the set introduced in Lemma 4.2 to block all back-door paths from X_{t-1} to W_{t-1} and W_t .



(c) A candidate FT-ADMG of the SCG in Figure 3b showing the set introduced in Lemma 4.4 to block all back-door paths from X_t to Y_t .



(b) A candidate FT-ADMG of the SCG in Figure 2g showing the set introduced in Lemma 4.3 to block all back-door paths from X_{t-1} to W_{t-1} and W_t .



(d) A candidate FT-ADMG of the SCG in Figure 2c showing the set introduced in Lemma 4.5 to block all back-door paths from W_{t-1} and W_t to Y_t .

Figure 5: Four FT-ADMGs that correspond respectively to the SCGs in Figures 2b, 2g 3b, and 2c. In each graph, the red and blue vertices represent the total effect of interest, while the green vertices are those that intercept all directed paths from the cause to the effect of interest. In FT-ADMGs (a), (b), and (c), all back-door paths from the red vertex (in bold) to the green vertices (in bold) are blocked by the brown, pink, and gray vertices. In FT-ADMG (d), all back-door paths from each green vertex (in bold) to the blue vertex (in bold) are blocked by the orange, purple, and other green vertices that are temporally prior to the selected green vertex.

Therefore, by Lemma 4.2, 4.3, and 4.4, \mathbb{B}^x statisfies the standard back-door criterion (Pearl, 1995) relative to $(X_{t-\gamma}, \mathbb{F})$ which means that by Pearl (1995, Theorem 1), $\Pr(\mathbb{F} \mid do(x_{t-\gamma})) = \sum_{\mathbb{B}^x} \Pr(\mathbb{F} \mid x_{t-\gamma}, \mathbb{B}^x) \Pr(\mathbb{B}^x)$. By Lemma 4.5, $\mathbb{B}^f \cup \{X_{t-\gamma}\}$ blocks all back-door paths from $W_{t-\lambda}$ to Y_t for all $0 \leq \lambda \leq \gamma$, except those passing by $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}\setminus\{W_{t-\lambda}\}$ and it does not contain any descendant of $W_{t-\lambda}$. Therefore, since it is intervened on $\mathbb{F} = \{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}, \mathbb{B}^f \cup \{X_{t-\gamma}\}$ satisfies the back-door criterion relative to (\mathbb{F}, Y_t) , which

means that by Pearl (1995, Theorem 1), $\Pr(y_t \mid do(\mathbb{f})) = \sum_{\mathbb{b}^f, x'_{t-\gamma}} \Pr(y_t \mid \mathbb{f}, \mathbb{b}^f, x'_{t-\gamma}) \Pr(\mathbb{b}^f).$

For simplicity, Theorem 1 is stated under a strict positivity condition on the distribution, which should ideally be relaxed, as discussed in Hwang et al. (2024). Furthermore, the sets used in Equation 1 of Theorem 1 are applicable whether Condition 4a or Condition 4b or Condition 4c of Definition 7 is statisfied. Nevertheless, by Lemma 4.2 the set \mathbb{B}^x in Theorem 1 can be reduced to $\{(B_{t-\gamma-\ell})_{0\leq\ell\leq\gamma_{\max}}|B\in Par(X,\mathcal{G}^s)\} \cup$ $\{(B_{t-\gamma-\ell})_{1\leq\ell\leq\gamma_{\max}}|B\in Anc(\mathbb{W},\mathcal{G}^s)\cap Desc(X,\mathcal{G}^s)\}\cup\{(X_{t-\gamma+\ell})_{1\leq\ell\leq\gamma}\}$ when Condition 4a is satisfied and it can be reduced to $\{(B_{t-\gamma-\ell})_{0\leq\ell<\gamma_{\max}}|B\in Par(X,\mathcal{G}^s)\}\cup\{(X_{t-\gamma-\ell})_{1\leq\ell<\gamma_{\max}}\}$ when Condition 4b is satisfied.

Definition 7 reveals a broad class of SCGs that enable the identification of the total effect using the do-free formula provided in Theorem 1. This significantly enhances the applicability of causal inference in realworld scenarios where the fully specified causal graph is unknown, but its SCG is available. For example, consider a conceptual study aimed at evaluating the effect of sedation levels administered today on blood pressure regulation tomorrow. Mathematically, this corresponds to estimating $Pr(BloodPressure_{tomorrow} =$ $b \mid Sedative_{today} = s$) for a given patient in critical care. Suppose daily patient monitoring records sedation levels, arterial pressure, and heart rate variability over time. Let us assume that the SCG corresponding to these time series is the one depicted in Figure 1c, where X represents sedation levels, Y corresponds to arterial pressure, and W denotes heart rate variability. In this setup, the effect of sedation on blood pressure is fully mediated by heart rate variability. Note that the SCG encodes the assumption of the presence of a latent confounder (physiological factor) between sedation needs and blood pressure regulation, and latent confounding between different time points of X and latent confounding between different timepoints of Y. In this scenario, the total effect of interest is not identifiable by adjustment alone, as the back-door criterion is not applicable. The standard front-door criterion and the Granger front-door criterion are not applicable for the reasons given in Section 3. However, the SCG-front-door criterion and the corresponding theorem provide the appropriate tools to correctly quantify the total effect of sedation levels administered today on blood pressure regulation tomorrow from non-experimental data, assuming all underlying assumptions hold. In this case, the appropriate do-free formula is given by:

$$\Pr(y_t \mid x_{t-1}) = \sum_{w_t, w_{t-1}} \sum_{x_{t-2}, x_t} \Pr(w_t, w_{t-1} \mid x_{t-1}, x_{t-2}, x_t)$$
$$\sum_{w_{t-2}, x_{t-2}, x_t, x'_{t-1}} \Pr(y_t \mid w_t, w_{t-1}, w_{t-2}, x_{t-2}, x_t, x'_{t-1}) \Pr(w_{t-2}, x_{t-2}, x_t, x'_{t-1}).$$

5 Towards non identifiability results

This section explains why the total effect $Pr(y_t \mid do(x_{t-\gamma}))$ is not identifiable in any of the SCGs shown in Figure 4.

Consider any SCG in Figure 4. $\Pr(y_t \mid do(x_{t-\gamma}))$ is not identifiable by the back-door criterion (Pearl, 1995) because the back-door path $X_{t-\gamma} \leftrightarrow Y_t$ cannot be blocked by any observed micro vertex. In the following, it is demonstrated that for each SCG, the total effect cannot be identified by decomposing it into other total effects and then multiplying them together.

Consider the SCG \mathcal{G}^s in Figure 4a. Let us show that if $X \leftrightarrows W$ in \mathcal{G}^s , then there exists at least one λ where $0 \leq \lambda < \gamma_{\max}$ such that $\Pr(w_{t-\lambda} \mid do(x_{t-\gamma}))$ is not identifiable. For $\gamma = \lambda$, there exists an FT-ADMG $\mathcal{G}_1 \in \mathcal{C}(\mathcal{G}^s)$ such that $X_t \in Par(W_t, \mathcal{G}_1)$ and another FT-ADMG $\mathcal{G}_2 \in \mathcal{C}(\mathcal{G}^s)$ where $W_t \in Par(X_t, \mathcal{G}_2)$. Thus, the total effect is not identifiable. The same logic can be applied to the SCGs in Figure 4c and a similar logic for Figure 4b to show that $\Pr(y_t \mid do(w_t))$ is not identifiable. Consider the SCG \mathcal{G}^s in Figure 4d. Let us demonstrate that if $X \to W \to U \to X$ exists in \mathcal{G}^s , then there exists at least one λ where $0 \leq \lambda < \gamma_{\max}$ such that $\Pr(w_{t-\lambda} \mid do(x_{t-\gamma}))$ is not identifiable. Let's take $\gamma = \lambda$. Since there is a back-door path $\pi^s = \langle X, U, W \rangle$, there exists an FT-ADMG \mathcal{G}_1 with the back-door path $\pi_1 = X_t \leftarrow U_t \leftarrow W_t$, which would need to be blocked by conditioning on U_t . However, there is also a directed path $\pi^s = \langle X, W, Z \rangle$, which corresponds to an FT-ADMG \mathcal{G}_2 with the directed path $\pi_2 = X_t \to W_t \to U_t$, where conditioning on U_t would bias the estimation of the total effect. This ambiguity around U_t implies that the total effect is not identifiable. A similar argument can be made for the SCG in Figure 4e, to show that $\Pr(y_t \mid do(w_t))$ is not identifiable.

6 Conclusion

This paper focuses on identifying total effects from summary causal graphs with latent confounding. Definition 7 establishes graphical conditions that are sufficient, under any underlying positive probability distribution, for the identifiability of the total effect. Additionally, in cases where identifiability is established, Theorem 1 provides a do-free formula for estimating the total effect. These results contribute to the ongoing effort to understand and estimate total effects from observational data using summary causal graphs. The main limitation of our result is that, while it is sound, it is not complete for identifying total effects using summary causal graphs. Consequently, future work should aim to develop a *complete* identifiability result that accounts for latent confounding and cycles, along with a corresponding do-free formula that does not require any information about the distribution.

Acknowledgments

I thank Fabrice Carrat, Nathanael Lapidus, and Benjamin Glemain for several discussions about applications of this work in epidemiology. This work was supported by the CIPHOD project (ANR-23-CPJ1-0212-01).

References

- Tara V. Anand, Adele H. Ribeiro, Jin Tian, and Elias Bareinboim. Causal effect identification in cluster dags. Proceedings of the AAAI Conference on Artificial Intelligence, 37(10):12172–12179, Jun. 2023.
- Charles K. Assaad, Emilie Devijver, and Eric Gaussier. Survey and evaluation of causal discovery methods for time series. J. Artif. Int. Res., 73, may 2022. ISSN 1076-9757. doi: 10.1613/jair.1.13428.
- Charles K. Assaad, Imad Ez-Zejjari, and Lei Zan. Root cause identification for collective anomalies in time series given an acyclic summary causal graph with loops. In Francisco Ruiz, Jennifer Dy, and Jan-Willem van de Meent (eds.), Proceedings of The 26th International Conference on Artificial Intelligence and Statistics, volume 206 of Proceedings of Machine Learning Research, pp. 8395–8404. PMLR, 25–27 Apr 2023.
- Charles K. Assaad, Emilie Devijver, Eric Gaussier, Gregor Goessler, and Anouar Meynaoui. Identifiability of total effects from abstractions of time series causal graphs. In *Proceedings of the Fortieth Conference on Uncertainty in Artificial Intelligence*, Proceedings of Machine Learning Research. PMLR, 2024.
- Ali Aït-Bachir, Charles K. Assaad, Christophe de Bignicourt, Emilie Devijver, Simon Ferreira, Eric Gaussier, Hosein Mohanna, and Lei Zan. Case studies of causal discovery from it monitoring time series. The History and Development of Search Methods for Causal Structure Workshop at the 39th Conference on Uncertainty in Artificial Intelligence, 2023.
- Gilles Blondel, Marta Arias, and Ricard Gavaldà. Identifiability and transportability in dynamic causal networks. *International Journal of Data Science and Analytics*, 3:131–147, 2016.
- Fabrice Carrat, Xavier de Lamballerie, Delphine Rahib, Hélène Blanché, Nathanael Lapidus, Fanny Artaud, Sofiane Kab, Adeline Renuy, Fabien Szabo de Edelenyi, Laurence Meyer, Nathalie Lydié, Marie-Aline Charles, Pierre-Yves Ancel, Florence Jusot, Alexandra Rouquette, Stéphane Priet, Paola Mariela Saba Villarroel, Toscane Fourié, Clovis Lusivika-Nzinga, Jérôme Nicol, Stephane Legot, Nathalie Druesne-Pecollo,

Younes Esseddik, Cindy Lai, Jean-Marie Gagliolo, Jean-François Deleuze, Nathalie Bajos, Gianluca Severi, Mathilde Touvier, Marie Zins, for the SAPRIS, and SAPRIS-SERO study groups. Antibody status and cumulative incidence of SARS-CoV-2 infection among adults in three regions of France following the first lockdown and associated risk factors: a multicohort study. *International Journal of Epidemiology*, 50 (5):1458–1472, 07 2021. ISSN 0300-5771. doi: 10.1093/ije/dyab110.

- Michael Eichler and Vanessa Didelez. Causal reasoning in graphical time series models. In Proceedings of the Twenty-Third Conference on Uncertainty in Artificial Intelligence, UAI'07, pp. 109–116, Arlington, Virginia, USA, 2007. AUAI Press. ISBN 0974903930.
- Michael Eichler and Vanessa Didelez. On granger-causality and the effect of interventions in time series. *Lifetime Data Analysis*, 16, 02 2010. doi: 10.1007/s10985-009-9143-3.
- Simon Ferreira and Charles K. Assaad. Identifiability of direct effects from summary causal graphs. Proceedings of the AAAI Conference on Artificial Intelligence, 38, 2024.
- Simon Ferreira and Charles K. Assaad. Identifying macro conditional independencies and macro total effects in summary causal graphs with latent confounding. *Proceedings of the AAAI Conference on Artificial Intelligence*, 39(25):26787–26795, Apr. 2025. doi: 10.1609/aaai.v39i25.34882.
- Isabel Fulcher, Ilya Shpitser, Stella Marealle, and Eric Tchetgen. Robust inference on population indirect causal effects: The generalized front door criterion. Journal of the Royal Statistical Society: Series B (Statistical Methodology), 82, 11 2019. doi: 10.1111/rssb.12345.
- Andreas Gerhardus and Jakob Runge. High-recall causal discovery for autocorrelated time series with latent confounders. In H. Larochelle, M. Ranzato, R. Hadsell, M.F. Balcan, and H. Lin (eds.), Advances in Neural Information Processing Systems, volume 33, pp. 12615–12625. Curran Associates, Inc., 2020.
- Benjamin Glemain, Xavier Lamballerie, Marie Zins, Gianluca Severi, Mathilde Touvier, Jean-François Deleuze, Fabrice Carrat, Pierre-Yves Ancel, Marie-Aline Charles, Sofiane Kab, Adeline Renuy, Stephane Le-Got, Celine Ribet, Mireille Pellicer, Emmanuel Wiernik, Marcel Goldberg, Fanny Artaud, Pascale Gerbouin-Rérolle, and Melody Enguix. Estimating sars-cov-2 infection probabilities with serological data and a bayesian mixture model. *Scientific Reports*, 14, 04 2024. doi: 10.1038/s41598-024-60060-3.
- Inwoo Hwang, Yesong Choe, Yeahoon Kwon, and Sanghack Lee. On positivity condition for causal inference. In *Forty-first International Conference on Machine Learning*, 2024.
- Kosukea Inoue, Beateb Ritz, and Onyebuchi A. Arah. Causal effect of chronic pain on mortality through opioid prescriptions: Application of the front-door formula. *Epidemiology (Cambridge, Mass.)*, 33:572–580, 2022. doi: 10.1097/EDE.000000000001490.
- Nathanael Lapidus, Juliette Paireau, Daniel Levy-Bruhl, Xavier de Lamballerie, Gianluca Severi, Mathilde Touvier, Marie Zins, Simon Cauchemez, and Fabrice Carrat. Do not neglect sars-cov-2 hospitalization and fatality risks in the middle-aged adult population. *Infectious Diseases Now*, 51(4):380–382, 2021. ISSN 2666-9919. doi: https://doi.org/10.1016/j.idnow.2020.12.007.
- Marloes Maathuis and Diego Colombo. A generalized backdoor criterion. The Annals of Statistics, 43, 07 2013. doi: 10.1214/14-AOS1295.
- Søren Wengel Mogensen, Daniel Malinsky, and Niels Richard Hansen. Causal learning for partially observed stochastic dynamical systems. In *Conference on Uncertainty in Artificial Intelligence*, 2018.
- Jerzy Neyman, D. M. Dabrowska, and T. P. Speed. On the Application of Probability Theory to Agricultural Experiments. Essay on Principles. Section 9. *Statistical Science*, 5(4):465 472, 1990. doi: 10.1214/ss/1177012031.
- Judea Pearl. Mediating instrumental variables. Technical Report R-210, Cognitive Systems Laboratory, UCLA Computer Science Department, 1993a.

- Judea Pearl. [Bayesian Analysis in Expert Systems]: Comment: Graphical Models, Causality and Intervention. *Statistical Science*, 8(3):266 – 269, 1993b. doi: 10.1214/ss/1177010894.
- Judea Pearl. Causal diagrams for empirical research. Biometrika, 82(4):669–688, 1995.
- Judea Pearl. The seven tools of causal inference, with reflections on machine learning. *Commun. ACM*, 62 (3):54–60, feb 2019. ISSN 0001-0782. doi: 10.1145/3241036.
- Judea Pearl et al. Models, reasoning and inference. Cambridge, UK: CambridgeUniversityPress, 19:2, 2000.
- Emilija Perkovic. Identifying causal effects in maximally oriented partially directed acyclic graphs. In Jonas Peters and David Sontag (eds.), Proceedings of the 36th Conference on Uncertainty in Artificial Intelligence (UAI), volume 124 of Proceedings of Machine Learning Research, pp. 530–539. PMLR, 03–06 Aug 2020.
- Jonas Peters, D. Janzing, and B. Schölkopf. Causal inference on time series using restricted structural equation models. In *Advances in Neural Information Processing Systems* 26, pp. 154–162, 2013.
- Marco Piccininni, Tobias Kurth, Heinrich Audebert, and Jessica Rohmann. The effect of mobile stroke unit care on functional outcomes: An application of the front-door formula. *Epidemiology (Cambridge, Mass.)*, 34:712–720, 06 2023. doi: 10.1097/EDE.000000000001642.
- Nicolas-Domenic Reiter, Jonas Wahl, Andreas Gerhardus, and Jakob Runge. Causal inference on process graphs, part ii: Causal structure and effect identification, 2024.
- Thomas Richardson. Markov properties for acyclic directed mixed graphs. *Scandinavian Journal of Statistics*, 30(1):145–157, 2003. ISSN 03036898, 14679469.
- R. W. Robinson. Counting unlabeled acyclic digraphs. In Charles H. C. Little (ed.), Combinatorial Mathematics V, pp. 28–43, Berlin, Heidelberg, 1977. Springer Berlin Heidelberg. ISBN 978-3-540-37020-8.
- Jakob Runge, Peer Nowack, Marlene Kretschmer, Seth Flaxman, and Dino Sejdinovic. Detecting and quantifying causal associations in large nonlinear time series datasets. *Science Advances*, 5(11):eaau4996, 2019. doi: 10.1126/sciadv.aau4996.
- Ilya Shpitser and Judea Pearl. Complete identification methods for the causal hierarchy. Journal of Machine Learning Research, 9:1941–1979, 2008.
- Ilya Shpitser, Tyler VanderWeele, and James M. Robins. On the validity of covariate adjustment for estimating causal effects. In *Proceedings of the Twenty-Sixth Conference on Uncertainty in Artificial Intelligence*, UAI'10, pp. 527–536, Arlington, Virginia, USA, 2010. AUAI Press. ISBN 9780974903965.
- Peter Spirtes, Clark N Glymour, Richard Scheines, and David Heckerman. *Causation, prediction, and search.* MIT press, 2000.
- Jin Tian and Judea Pearl. On the testable implications of causal models with hidden variables. In Proceedings of the Eighteenth Conference on Uncertainty in Artificial Intelligence, UAI'02, pp. 519–527, San Francisco, CA, USA, 2002. Morgan Kaufmann Publishers Inc. ISBN 1558608974.
- Jonas Wahl, Urmi Ninad, and Jakob Runge. Foundations of causal discovery on groups of variables. *Journal of Causal Inference*, 12(1):20230041, 2024. doi: doi:10.1515/jci-2023-0041.
- Tian-Zuo Wang, Tian Qin, and Zhi-Hua Zhou. Estimating possible causal effects with latent variables via adjustment. In Andreas Krause, Emma Brunskill, Kyunghyun Cho, Barbara Engelhardt, Sivan Sabato, and Jonathan Scarlett (eds.), Proceedings of the 40th International Conference on Machine Learning, volume 202 of Proceedings of Machine Learning Research, pp. 36308–36335. PMLR, 23–29 Jul 2023.
- Ryan Wiegand, Yangyang Deng, Xiaoyi Deng, Adam Lee, William Meyer, Stanley Letovsky, Myrna Charles, Adi Gundlapalli, Adam MacNeil, Aron Hall, Natalie Thornburg, Jefferson Jones, Ronaldo Iachan, and Kristie Clarke. Estimated sars-cov-2 antibody seroprevalence trends and relationship to reported case prevalence from a repeated, cross-sectional study in the 50 states and the district of columbia, united states—october 25, 2020–february 26, 2022. The Lancet Regional Health - Americas, 18:100403, 02 2023. doi: 10.1016/j.lana.2022.100403.

A Appendix

Lemma 4.1. Consider an SCG \mathcal{G}^s . If a set of macro vertices \mathbb{W} intercepts all directed paths from X to Y in \mathcal{G}^s then $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}$ intercepts all directed paths from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Proof. Let \mathcal{G} be a candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$. Consider a directed path π from $X_{t-\gamma}$ to Y_t in \mathcal{G} . Suppose π does not contain any vertex from $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}$. Then, π must include at least one vertex from $\{(\mathbb{W}_{t-\gamma+\ell})_{\ell\in[t_0,t_{max}]\setminus\{0,\cdots,\gamma\}}\}$, as all paths from X to Y are intercepted by \mathbb{W} in \mathcal{G}^s . Consider the case where there exists $W_{t-\gamma+\ell}$ in π such that $\ell < 0$. In this case, notice that $W_{t-\gamma+\ell}$ is temporally prior to $X_{t-\gamma}$ which means that by Assumption 1, π cannot be a directed path because there exists no directed path from $X_{t-\gamma}$ to $W_{t-\gamma+\ell}$. Consider the case where there exists $W_{t-\gamma+\ell}$ in π such that $\ell > \gamma$. In this case, notice that Y_t is temporally prior to $W_{t-\gamma+\ell}$ which means that by Assumption 1, π cannot be a directed path because there exists no directed path because there exists no directed path from $W_{t-\gamma+\ell}$ to Y_t . Therefore, it must be the case that π includes at least one vertex from $(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}$. Since \mathcal{G} and π are arbitrary, this conclusion applies to all directed paths between from $X_{t-\gamma}$ to Y_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Property 1. Consider an SCG \mathcal{G}^s . If there is no backdoor path from X to W in \mathcal{G} , then each active backdoor paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ in any FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$ has to pass by another vertex $X_{t-\lambda'}$ with $\lambda' \neq \gamma$.

Proof. Consider an SCG \mathcal{G}^s and a given candidate FT-ADMG \mathcal{G} in $\mathcal{C}(\mathcal{G}^s)$. First, note that a backdoor path from $X_{t-\gamma}$ to Y_t in \mathcal{G} cannot consist solely of edges pointing toward $X_{t-\gamma}$ because if this were the case, there would be a backdoor from X to W path in \mathcal{G}^s . This would contradict the assumption that there is no backdoor path from X to W. Therefore, the only possible backdoor path, is a path containing at least one ancestor of $X_{t-\gamma}$ and $W_{t-\lambda}$. Consider an active backdoor path π in \mathcal{G} : $X_{t-\gamma} \leftarrow U_{t'}, \cdots, U_{t''}, \cdots \leftarrow$ $S_{t'''} \cdots, Q_{t''''} \rightarrow \cdots, Z_{t'''''} \rightarrow W_{t-\lambda}$. Let's assume that there exists no $X_{t-\lambda'}$ such that $\lambda' \neq \gamma$ that belongs to π . Since the path π does not pass through any instance of X other than $X_{t-\gamma}$, the path that is compatible with π in \mathcal{G}^s is the following path : $X \leftarrow U \cdots \leftarrow S \cdots Q \rightarrow \cdots, Z \rightarrow W$ where each subpath of micro vertices with the left and right endpoints shares the same macro vertex are replaced by this macro vertex. The path we've identified corresponds directly to a backdoor path in \mathcal{G}^s . This contradiction implies that the assumption we took must be false. Moreover, if we assume that $X_{t-\lambda'}$ belongs to the path then if we replace $Q_{t''''}$ by $X_{t-\lambda'}$ then the compatible path in \mathcal{G}^s would be $X \to Z \to W$ which is not a backdoor path. \Box

Lemma 4.2. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. If $Cycles(X, \mathcal{G}^s) = \emptyset$ then for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(B_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma+\ell})_{1 \leq \ell \leq \gamma}\}$ blocks all back-door paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ and does not contain any descendant of $X_{t-\gamma}$ in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Case 1 If $\lambda' < 0$ then obviously, by Assumption 1, the backdoor path is necessarily blocked by the empty set since $X_{t-\lambda'}$ cannot be an ancestor of $W_{t-\lambda}$ nor an ancestor of $X_{t-\gamma}$.

Case 2 If $\lambda' > \gamma$, then by Condition 2, $\{(X_{t-\gamma-\ell})_{\ell \leq 1}\}$ can blocks the backdoor path, however this set is unbounded and therefore unpratical. So instead, we can replace it by $\{(B_{t-\gamma-\ell})_{1\leq \ell \leq \gamma_{\max}}|B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G}^s)\}$ (remark that $\{(X_{t-\gamma-\ell})_{1\leq \ell \leq \gamma_{\max}}\}$ is included in that set) to blocks the backdoor path. Furthermore, by Assumption 1, $\{(B_{t-\gamma-\ell})_{1\leq\ell\leq\gamma_{\max}}|B\in Anc(\mathbb{W},\mathcal{G}^s)\cap Desc(X,\mathcal{G}^s)\}$ cannot contain descendants of $X_{t-\gamma}$.

Case 3 If $0 \le \lambda' < \gamma$, then by Condition 2, $\{(X_{t-\gamma+\ell})_{1 \le \ell \le \gamma}\}$ blocks the backdoor path. Furthermore, since $Cycles(X, \mathcal{G}^s) = 0, \{(X_{t-\gamma+\ell})_{1 \le \ell \le \gamma}\}$ cannot contain descendants of $X_{t-\gamma}$.

Lemma 4.3. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. If $Cycle(X, \mathcal{G}^s) = \{X \to X\}$ and $|\exists Z \in Anc(X, \mathcal{G}^s)$ such that $X \leftrightarrow Z$ in \mathcal{G}^s then for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \leq \ell \leq \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{\max}}\}$ blocks all back-door paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ and does not contain any descendant of $X_{t-\gamma}$ in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Proof. Condition 2 implies that there exists no active backdoor path $\pi^s = \langle V^1 = X, \ldots, V^n = W \rangle$ from X to W in \mathcal{G}^s such that $\langle V^2, \ldots, V^{n-1} \rangle \subseteq Desc(X, \mathcal{G}^s)$ which means that there are no cycles that contain X and another vertex on a path from X to W. Furthermore, by Property 1, all backdoor paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ in any FT-ADMG \mathcal{G} in $\mathcal{C}(\mathcal{G}^s)$ has to pass by another vertex $X_{t-\lambda'}$ with $\lambda' \neq < \gamma$. Let us consider three cases:

- Case 1 If $\lambda' < 0$, then obviously, by Assumption 1, the backdoor path is necessarily blocked by the empty set since $X_{t-\lambda'}$ cannot be an ancestor of $W_{t-\lambda}$ nor an ancestor of $X_{t-\gamma}$.
- Case 2 If $\lambda' \geq \gamma$ and if the second micro vertex $U_{t-\lambda''}$ on the backdoor path we are considering is compatible with the macro vertex $U \notin Cycles(X, \mathcal{G}^s)$ then the backdoor path can be blocked by $\{(B_{t-\gamma-\ell})_{0\leq\ell\leq\gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\}.$
- Case 3 If $\lambda' \geq \gamma$ and if the second micro vertex $U_{t-\lambda''}$ on the backdoor path we are considering is compatible with the macro vertex $U \in Cycles(X, \mathcal{G}^s)$ then, since we consider that $Cycle(X, \mathcal{G}^s) = \{X \to X\}$, U = X and $\lambda' > \gamma$. Therefore, the backdoor path starts with $X_{t-\gamma} \leftarrow X_{t-\lambda'}$ which means it is obviously blocked by $\{(X_{t-\gamma-\ell})_{1 \leq \ell \leq \gamma_{\max}}\}$.

In addition, by construction and by Assumption 1, $\{(B_{t-\ell})_{0 \le \ell \le \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma-\ell})_{1 \le \ell \le \gamma_{\max}}\}$ cannot contain any descendant of $X_{t-\gamma}$. Which means that $\{(B_{t-\ell})_{0 \le \ell \le \gamma_{\max}} | B \in Par(X, \mathcal{G}^s)\} \cup \{(X_{t-\gamma-\ell})_{1 \le \ell \le \gamma_{\max}}\}$ blocks all backdoor paths from $X_{t-\gamma}$ to $W_{t-\lambda}$ starting with an undashed right arrow, i.e, " $X_{t-\gamma} \leftarrow$ ". Since we consider that $\not\exists Z \in Anc(X, \mathcal{G}^s)$ such that $X \leftrightarrow Z$, no additional back-door paths can exist from $X_{t-\gamma}$ to $W_{t-\lambda}$.

Lemma 4.4. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices (X_t, Y_t) compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices (X_t, Y_t) . Then for any $W_t \in \mathbb{W}_t$, the set $\{(B_{t-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Anc(X, \mathcal{G}^s) \setminus Desc(X, \mathcal{G})\} \cup \{(B_{t-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in (Anc(X, \mathcal{G}^s) \cup Anc(\mathbb{W}, \mathcal{G}^s)) \cap Desc(X, \mathcal{G}^s)\}$ blocks all back-door paths from X_t to W_t and does not contain any descendant of X_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$,

Proof. Condition 2 implies that there exists no active backdoor path $\pi^s = \langle V^1 = X, \dots, V^n = W \rangle$ from X to W in \mathcal{G}^s such that $\langle V^2, \dots, V^{n-1} \rangle \subseteq Desc(X, \mathcal{G}^s)$ which means that there are no cycles that contain X and another vertex on a path from X to W. Furthermore, by Property 1, all backdoor paths from X_t to W_t in any FT-ADMG \mathcal{G} in $\mathcal{C}(\mathcal{G}^s)$ has to pass by another vertex $X_{t-\lambda'}$ with $\lambda' \neq 0$. Let us consider three cases:

- Case 1 If $\lambda' < 0$, then obviously, by Assumption 1, the backdoor path is necessarily blocked by the empty set since $X_{t-\lambda'}$ cannot be an ancestor of W_t nor an ancestor of X_t .
- Case 2 If $\lambda' > 0$ and if the second micro vertex $U_{t-\lambda}$ on the backdoor path we are considering is compatible with the macro vertex $U \notin Cycles(X, \mathcal{G}^s)$ then the backdoor path can be blocked by $\{(B_{t-\ell})_{0 \le \ell \le \gamma_{\max}} | B \in Anc(X, \mathcal{G}^s) \setminus Desc(X, \mathcal{G})\}.$

Case 3 If $\lambda' > 0$ and if the second micro vertex $U_{t-\lambda}$ on the backdoor path we are considering is compatible with the macro vertex $U \in Cycles(X, \mathcal{G}^s)$ then the backdoor path can be blocked by $\{(B_{t-\ell})_{0 \leq \ell \leq \gamma_{\max}} | B \in Anc(X, \mathcal{G}^s) \setminus Desc(X, \mathcal{G})\} \cup \{(B_{t-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in Anc(X, \mathcal{G}^s) \cap Desc(X, \mathcal{G})\}$.

- Case 1 If $\lambda' < 0$, then obviously, by Assumption 1, the backdoor path is necessarily blocked by the empty set since $X_{t-\lambda'}$ cannot be an ancestor of W_t nor an ancestor of X_t .
- Case 2 If $\lambda' > 0$, then by Condition 2, $\{(X_{t-\gamma-\ell})_{\ell \leq 1}\}$ can blocks the backdoor path, however this set is unbounded and therefore unpratical. So instead, we can replace it by $\{(B_{t-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G})\}$. Furthermore, by Assumption 1, $\{(B_{t-\ell})_{1 \leq \ell \leq \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(X, \mathcal{G})\}$ cannot contain any descendant of X_t in any candidate FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$.

Property 2. Consider an SCG \mathcal{G}^s . Given Condition 3, if all backdoor path from W to Y in \mathcal{G} are blocked by X, then each active backdoor paths from $W_{t-\lambda}$ to Y_t in any FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$ has to pass either by a vertex $X_{t-\lambda'}$ or a vertex $W_{t-\lambda''}$ with $\lambda'' \neq \lambda$.

Proof. Consider an SCG \mathcal{G}^s and a given candidate FT-ADMG \mathcal{G} in $\mathcal{C}(\mathcal{G}^s)$. The proof that there might exists an active backdoor path from $W_{t-\lambda}$ to Y_t that does not pass through any $X_{t-\lambda'}$ must pass through a vertex $W_{t-\lambda''}$ follows a similar reasoning to that used in the proof of Property 1. Let's assume there is an active backdoor path between $W_{t-\lambda}$ and Y_t that does not pass through $W_{t-\lambda''}$. If the path between $W_{t-\lambda}$ and Y_t is active and does not pass through $W_{t-\lambda'}$, then this path must be relying on connections in the SCG. Consider such an active backdoor path in \mathcal{G} : $W_{t-\lambda} \leftarrow U_{t'}, \cdots, U_{t''}, \cdots \leftarrow S_{t'''} \cdots, Q_{t''''} \rightarrow \cdots, Z_{t'''''} \rightarrow Y_t$. The compatible of the path considered in \mathcal{G}^s is the following path : $W \leftarrow U \cdots \leftarrow S \cdots Q \rightarrow \cdots, Z \rightarrow Y$ where each subpath of micro vertices with the left and right endpoints shares the same macro vertex are replaced by this macro vertex. The identified path corresponds to a backdoor path in \mathcal{G}^s . If X is not on this implies that our initial assumption—that there exists a backdoor path between $W_{t-\lambda}$ and Y_t that does not pass through $W_{t-\lambda''}$ —must be false. Consequently, under our assumption, X must be on the path in \mathcal{G}^s , and $X_{t-\lambda'}$ must be on the corresponding path in \mathcal{G} .

Now, under a similar scenario, let us assume that the backdoor path between $W_{t-\lambda}$ and Y_t does pass through $W_{t-\lambda''}$. For instance, suppose $W_{t-\lambda''} = Q_{t'''}$ on this path. In this case, the compatible path in \mathcal{G}^s would be $W \to \cdots \to Z \to Y$, which is not a backdoor path in \mathcal{G}^s .

Lemma 4.5. Consider an SCG $\mathcal{G}^s = (\mathbb{S}, \mathbb{E}^s)$ and the pair of micro vertices $(X_{t-\gamma}, Y_t)$ compatible with the macro vertices (X, Y). Suppose \mathbb{W} is a set of macro vertices that statisfies Conditions 1, 2 and 3 of Definition 7 in \mathcal{G}^s relative to the pair of micro vertices $(X_{t-\gamma}, Y_t)$. Then, for any $W_{t-\lambda} \in \{(\mathbb{W}_{t-\gamma+\ell})_{0 \le \ell \le \gamma}\}$, the set $\{(B_{t-\gamma-\ell})_{-\gamma \le \ell \le \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \setminus Desc(\mathbb{W}, \mathcal{G}^s)\} \cup \{(B_{t-\gamma+\ell})_{1 \le \ell \le \gamma_{\max}} | B \in Anc(\mathbb{W}, \mathcal{G}^s) \cap Desc(\mathbb{W}, \mathcal{G}^s)\}$ blocks all back-door paths from $W_{t-\lambda}$ to Y_t not passing by $\{(\mathbb{W}_{t-\gamma+\ell})_{0 \le \ell \le \gamma}\} \setminus \{W_{t-\lambda}\}$ and it does not contain any descendent of $W_{t-\lambda}$.

Proof. If there is a cycle that includes W and any other vertex not in W on a directed path from X to Y that is intercepted by W and that is closer than W to Y on the path, then there exists an active backdoor path from W to Y that does not pass through X (which means cannot be blocked by X). By Conditions 2 and 3, this cannot be true. Therefore, it must be the case that all cycles containing W do not include any other vertex on a directed path from X to Y that are neither in W nor closer to Y on the path. By Property 2, all backdoor path from $W_{t-\lambda}$ to Y_t in any FT-ADMG in $\mathcal{C}(\mathcal{G}^s)$ has to pass by either by $X_{t-\lambda'}$ or by $W_{t-\lambda''}$ with $\lambda'' \neq \lambda$. Let's consider one of these backdoor paths that pass by a vertex $X_{t-\lambda'}$ but does not pass by any vertex $W_{t-\lambda''}$ such that $\lambda'' \neq \lambda$. By Condition 3, $\{(B_{t-\gamma-\ell})_{-\gamma \leq \ell \leq \gamma_{\max}} | B \in Anc(W, \mathcal{G}^s) \setminus Desc(W, \mathcal{G}^s)\}$ block this path, however it can activate a path of the form $W_{t-\lambda} \leftarrow \dots X_{t-\lambda'} \longleftrightarrow Y_t$ where $\lambda' > \gamma + \gamma_{\max}$. Since $\lambda \leq \gamma$ and $\lambda' > \gamma + \gamma_{\max}$ then the parents of $W_{t-\lambda}$ which are temporally prior to $t - \lambda$ can block this path. Notice that these parents are in $\{(B_{t-\gamma-\ell})_{1\leq \ell \leq \gamma_{\max}} | B \in Anc(W, \mathcal{G}^s) \cap Desc(W, \mathcal{G}^s)\}$. Furthermore, by construction and by Assumption 1, $\{(B_{t-\gamma-\ell})_{-\gamma \leq \ell \leq \gamma_{\max}} | B \in Anc(W, \mathcal{G}^s) \setminus Desc(W, \mathcal{G}^s)\} \cup \{(B_{t-\gamma-\ell})_{1\leq \ell \leq \gamma_{\max}} | B \in Anc(W, \mathcal{G}^s) \cap Desc(W, \mathcal{G}^s)\}$ does not contain any descendant of $W_{t-\lambda}$.

Now let's consider one of the remaining backdoor paths that pass by a vertex $W_{t-\lambda''}$. Let us consider two cases:

- Case 1 If $\lambda'' > \gamma$, then obviously the backdoor path can be blocked by $\{(B_{t-\gamma-\ell})_{1 \le \ell \le \gamma_{\max}} | B \in Anc(W, \mathcal{G}^s) \cap Desc(W, \mathcal{G}^s)\}.$
- Case 2 If $\lambda'' \leq \gamma$ then the backdoor path pass by $\{(\mathbb{W}_{t-\gamma+\ell})_{0\leq\ell\leq\gamma}\}\setminus\{W_{t-\lambda}\}$ which means it does not need to be blocked.